Author's response to reviews

Title: Relevance of circulating nucleosomes and oncological biomarkers for predicting response to transarterial chemoembolization therapy in liver cancer patients

Authors:

Nikolaus Kohles (nikolaus.kohles@med.uni-muenchen.de)
Dorothea Nagel (Dorothea.Nagel@med.uni-muenchen.de)
Dietrich Jüngst (Dietrich.Juengst@med.uni-muenchen.de)
Jürgen Durner (Juergen.Durner@med.uni-muenchen.de)
Petra Stieber (Petra.Stieber@med.uni-muenchen.de)
Stefan Holdenrieder (Stefan.Holdenrieder@med.uni-muenchen.de)

Version: 3 Date: 27 April 2011

Author's response to reviews:

Dear editors,

thank you very much for your kind letter and the positive responses of the reviewers.

The manuscript has now completely been revised by a native speaking scientist with long experience in copyediting scientific manuscripts. In addition we have performed thoroughful formatting of the reference section. All changes were done using the "track changes" function, so you easily will see the corrections and changes made.

We hope that the style of the written English meets your expectations and would be pleased if the manuscript is accepted for publication in BMC cancer now.

With best regards,

Nikolaus Kohles
Stefan Holdenrieder